Author's response to reviews

Title: Antioxidant and vasorelaxant activities induced by northeastern Brazilian fermented grape skins

Authors:

José George Albuquerque (jgeorge_farm@msn.com)
Valéria Lopes (val_farm@hotmial.com)
Artur José Almeida (arthur-jp@hotmail.com)
Ionaldo Diniz (ionaldobasilio@gmail.com)
Melissa Luciano (melissanegroluciano@hotmail.com)
Bruno Meireles (bruno_meireles7@hotmail.com)
Angela Cordeiro (atribuzycordeiro@gmail.com)
Islânia Araújo (islania.ltf@gmail.com)
Robson veras (robveras@msn.com.br)
Thaís Ribeiro (thaisprieiro@hotmail.com)
Isac Medeiros (isacmed@uol.com.br)

Version: 2 Date: 11 Jul 2017

Author's response to reviews:

- The email addresses of all manuscript authors were included on the manuscript’s Title Page.
- ARRIVE checklist and response to the reviewers were removed from the additional files.